Author's response to reviews

Title: Potential Angiogenic Effect Of Human Chorionic Gonadotropin Hormone In Rapid Progression And Spontaneous Regression Of Metastatic Renal Cell Carcinoma During Pregnancy And After Surgical Abortion. A case report.

Authors:

László Mangel (mangellaszlo@gmail.com)
Krisztina Bíró (szbirokriszta@gmail.com)
István Battyáni (batyo@chello.hu)
Péter Gőcze (peter.gocze@aok.pte.hu)
Tamás Tornóczky (ttamas64@hotmail.com)
Endre Kálmán (ke6100@gmail.com)

Version: 6 Date: 11 December 2015

Author's response to reviews: see over
Dear Editorial Office,

We send a brief answer to the reviewers’ comments below and to we would like to re-submit the revised MS to BMC Cancer with the title of "Potential angiogenic effect of hCG in rapid progression and spontaneous regression of metastatic renal cell carcinoma during pregnancy and after surgical abortion. A case report." to be considered for publication.

We would like to thank the positive opinion of the Reviewers once more. We tried to prepare the MS based on the final suggestions of Reviewer2: 1. We condensed the histology report to two third of the original length. 2. We inserted the details of the whole brain RT 3. We also completed the RT doses of other irradiation courses (page 6) 4. We inserted some more medical information based on CARE checklist, however we did not establish a time-table.

With best regards, László Mangel, Pécs, Hungary

Corresponding author: László Csaba Mangel MD, PhD
Institute of Oncotherapy, University of Pécs, Hungary,
Official address: H-7624, Édesanyák útja 17., Pécs, Hungary
Official phone: 36-72-536-080, cell phone: 36-20-340-9889, Fax number: 36-72-536-481
E-mail: mangellaszlo@gmail.com, mangel.laszlo@pte.hu